Author’s response to reviews

Title: Antimicrobial activities of endophytic fungi isolated from Ophiopogon japonicus (Liliaceae)

Authors:

Hanqiao Liang (keliaopeituola@yahoo.com.cn)
Yongmei Xing (meimary@163.com)
Juan Chen (kibchenjuan@126.com)
Dawei Zhang (zhangdw517@gmail.com)
Chunlan Wang (wangchunlan2006@163.com)
Shunxing Guo (sxquo2006@yahoo.com.cn)

Version: 4 Date: 14 August 2012

Author’s response to reviews: see over
Dear Dr. Rowles,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled “Antimicrobial activities of endophytic fungi isolated from *Ophiopogon japonicus*”. (ID: 2055995522732397).

We have studied reviewer’s comments carefully and have made revision which marked in red in the paper. We have tried our best to revise our manuscript according to the comments. Attached please find the revised version, which we would like to submit for your kind consideration.

We would like to express our great appreciation to you and reviewers for comments on our paper.

Looking forward to hearing from you.

Thank you and best regards.

Yours sincerely,

Dr. Liang

Corresponding author:

Name: Chunlan Wang   E-mail: wangchunlan2006@163.com

Shunxing Guo   E-mail: sxguo2006@yahoo.com.cn